Author's response to reviews

Title: Scale Development: Factors Affecting Diet, Exercise, and Stress Management (FADESM)

Authors:

Mei-Wei Chang (changme@msu.edu)
Roger Brown (rlbrown3@wiscmail.wisc.edu)
Susan Nitzke (nitzke@nutrisci.wisc.edu)

Version: 5 Date: 22 February 2008

Author's response to reviews: see over
Dr. Annabel Phillips, Senior Assistant Editor  
BMC-series Journals  
BioMed Central

Dear Dr. Phillips

We are submitting a revised manuscript entitled “Scale Development: Factors Affecting Diet, Exercise, and Stress Management (FADESM).” We have made formatting changes as requested. These comments are in bold type. Thanks for reviewing and considering this manuscript.

Table 1 column headings: the second set of column heading was removed.  
Figure title: the figure title was removed and was placed after references.  
Figure cropping: done  
Additional files: a section titled “Additional files” was added.

Authors: Mei-Wei Chang, PhD, RN; Roger Brown, PhD, Susan Nitzke, PhD, RD

Sincerely,

Mei-Wei Chang, PhD, RN  
Assistant Professor  
College of Nursing  
Michigan State University  
B515 F West Fee Hall  
East Lansing, MI 48824-1317  
Phone: 517-353-8682  
Fax: 517-353-8536  
Email: changme@msu.edu